Author's response to reviews

Title: Rationale and design of ASPIRE-ICU: a prospective cohort study on the incidence and predictors of Staphylococcus aureus and Pseudomonas aeruginosa pneumonia in the ICU

Authors:

Fleur Paling (f.p.paling@umcutrecht.nl)
Darren Troeman (D.P.R.Troeman@umcutrecht.nl)
Martin Wolkewitz (wolke@imbi.uni-freiburg.de)
Rubana Kalyani (kalyanirub@MedImmune.com)
Daniël Prins (D.R.Prins-2@umcutrecht.nl)
Susanne Weber (sweber@imbi.uni-freiburg.de)
Christine Lammens (christine.lammens@uantwerpen.be)
Leen Timbermont (Leen.Timbermont@uantwerpen.be)
Herman Goossens (Herman.Goossens@uza.be)
Surbhi Malhotra-Kumar (surbhi.malhotra@uantwerpen.be)
Frangiscos Sifakis (Frangiscos.Sifakis@astrazeneca.com)
Marc Bonten (M.J.M.Bonten@umcutrecht.nl)
Jan Kluytmans (jankluytmans@gmail.com)

Version: 2 Date: 12 Sep 2017

Author’s response to reviews:

1. Please remove your cover letter as this is no longer needed

The coverletter was removed.

2. The additional file containing the full list of ethics committees that approved this study & participating sites was not present. Please ensure this is added.

I added both documents to the online submission.
3. In your 'Site selection' could you update the text as at present it implies you have not chosen the sites.

Text was updated. I included a track-changed version as well as a clean version of the new manuscript.

4. Please include figure legends for your additional files in your main manuscript text.

I included the legend of the Table in the main manuscript text, and removed it from the Table itself. There are no (other) figures or tables with legends in the additional files.